Author’s response to reviews

Title: Removal of an endometrioma passing through the left greater sciatic foramen using a concomitant laparoscopic and transgluteal approach: case report

Authors:

Yanchun Liang (lychun1988@foxmail.com)
Yunhe Zhao (522158244@qq.com)
Meng Xia (xiameng25@163.com)
Shuqin Chen (chenshuqin1021@163.com)
Qingtang Zhu (zhuqingt@mail.sysu.edu.cn)
Shuzhong Yao (yszlfy@163.com)

Version: 1 Date: 20 Jun 2019

Author’s response to reviews:

Department of Obstetrics and Gynecology
The First Affiliated Hospital, Sun Yat-Sen University
No. 58, the 2nd Zhongshan Road
Yuexiu District
Guangzhou, 510080
Guangdong Province
China
20 Jun, 2019

Dear Robin L. Cassady-Cain, Ph.D.,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive
comments and suggestions on our manuscript entitled “Removal of an endometrioma across the left greater sciatic foramen using a concomitant laparoscopic and transgluteal approach" (BMWH-D-18-00662).

We have studied reviewer’s comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

1. Editor Comments:

   Please copyedit your manuscript to improve the standard of written English. If this is not possible, please consider using a professional service.

   Response: we have used professional language editing service, and relevant certification is attached in the annex.

2. Ferrero Simone (Reviewer 1):

   The first paragraph of the Discussion is not relevant and it should be deleted "Endometriosis is one of the most common benign gynecologic diseases in women of reproductive age, the morbidity of which reaches up to 10% [1]. The true frequency is actually …"

   Response: we deleted the first paragraph of the discussion.

We have tried our best to revise our manuscript according to the comments. Attached please find the revised version, which we would like to submit for your kind consideration.

Thank you very much for your considering our manuscript for potential publication. I'm looking forward to hearing from you soon.

Sincerely yours,

Shuzhong Yao